Author's response to reviews

Title: Measuring the health systems impact of disease control programs: Critical assessment of the WHO Building Blocks Framework

Authors:

Sandra Mounier-Jack (sandra.mounier-jack@lshtm.ac.uk)
Ulla K Griffiths (ulla.griffiths@lshtm.ac.uk)
Svea Closser (sclosser@gmail.com)
helen Burchett (helen.burchett@lshtm.ac.uk)
bruno Marchal (bmarchal@itg.be)

Version: 2 Date: 17 October 2013

Author's response to reviews: see over
Revision manuscript: 1917423906110836

Measuring the health systems impact of disease control programs: Critical assessment of the WHO Building Blocks Framework

The following changes have been made on the manuscript

- A ‘Competing interests’ section was included in the manuscript
- The Authors’ Contributions section was added
- The Authors’ information was moved after the Competing interests section.
- Acknowledgements were added